Author’s response to reviews

Title: Randomised controlled trial to improve health and reduce substance use in established psychosis (IMPaCT): Cost-effectiveness of integrated psychosocial health promotion

Authors:
Margaret Heslin (margaret.heslin@kcl.ac.uk)
Anita Patel (anita.patel@qmul.ac.uk)
Daniel Stahl (daniel.r.stahl@kcl.ac.uk)
Poonam Gardner-Sood (poonam.sood@kcl.ac.uk)
Manyara Mushore (mushorem@lsbu.ac.uk)
Shubulade Smith (shubulade.smith@kcl.ac.uk)
Kathy Greenwood (k.e.greenwood@sussex.ac.uk)
Oluwadamilola Onagbesan (Oluwadamilola.Onagbesan@kcl.ac.uk)
Conan O’Brien (conan.obrien@kcl.ac.uk)
Catherine Fung (Catherine.Fung@elft.nhs.uk)
Ruth Ohlsen (ruth.ohlsen@kcl.ac.uk)
David Hopkins (dhopkins3@nhs.net)
Philippa Lowe (lowephilippa@aol.com)
Maurice Arbutnot (mauricea@easy.com)
Stanley Mutsatsa (Stanley.mutsatsa@city.ac.uk)
Gill Todd (gillian.todd27@btinternet.com)
Anna Kolliakou (anna.Kolliakou@kcl.ac.uk)
John Lally (john.lally@kcl.ac.uk)
Brendon Stubbs (brendon.stubbs@kcl.ac.uk)
Khalida Ismail (khalida.2.ismail@kcl.ac.uk)
Anthony David (anthony.david@kcl.ac.uk)

Robin Murray (robin.murray@kcl.ac.uk)

Zerrin Atakan (zerrin.atakan@kcl.ac.uk)

Fiona Gaughran (fiona.p.gaughran@kcl.ac.uk)

**Version: 3 Date: 30 Aug 2017**

**Author’s response to reviews:**

The author contributions have been edited to:

AP, DS, SS, KG, DH, GT, KI, AD, RM, ZA and FG conceived and designed the study. MH conducted all data analysis with oversight from AP. PGS, MM, OO, COB, CF, RO, PL, MA, SM, AK, JL and BS all made substantial contributions to acquisition. MH, AP, DS, SS, KG, DH, GT, KI, AD, RM, ZA, FG PGS, MM, OO, COB, CF, RO, PL, MA, SM, AK, JL and BS all made substantial contributions to interpretation of data. MH and AP drafted the original manuscript. All authors were involved in critically revising the manuscript for important intellectual content, gave final approval of the published version and are accountable for all aspects of the work.

Sorry for the confusion, I hope this is correct now.